Author’s response to reviews

Title: The Hypocholesterolemic Activity of Thai Glutinous Rice (Oryza sativa L. var. Niaw San-pah-tawng) Fermented by Monascus purpureus in High-Fat Cholesterol Diet Fed Rats

Authors:

Kanokporn Saenphet (stit.lilo123@gmail.com)
Anurak Bunnoy (anurak.bunnoy@gmail.com)
Saisamorn Lumyong (scboi009@chaingmai.ac.th)
Supap Saenphet (bigsu.cmu@gmail.com)

Version: 4
Date: 8 October 2014

Author’s response to reviews: see over
Dear editor,

We are pleased to submit the revised manuscript (MS: 1037420545141680) entitled “The Hypocholesterolemic Activity of Thai Glutinous Rice (*Oryza sativa* L. var. Niaw San-pah-tawng) Fermented by *Monascus purpureus* in High-Fat Cholesterol Diet Fed Rats” by Kanokporn Saenphet, Anurak Bunnoy, Saisamorn Lumyong, Supap Saenphet and Siriwadee Chomdej that has been made for the editorial assessment.

The detail of rice identification is already added in the methodology section.

I would very appreciate if you could kindly consider my manuscript for the peer review process. Thank you very much in advance for your kind consideration.

Sincerely yours,

Assist. Prof. Dr. Kanokporn Saenphet

Department of Biology, Faculty of Science, Chiang Mai University, 239 Huay Kaew Road, Tumbol Suthep, Muang District, Chiang Mai, 50200, Thailand.
Phone: +66-86-605-0743, Fax: +66-53-892259